Author’s response to reviews

Title: Status of prehospital delay and intravenous thrombolysis in the management of acute ischemic stroke in Nepal

Authors:

Gaurav Nepal (gauravnepal@iom.edu.np)

Jayant Yadav (jkyadav22@iom.edu.np)

Babin Basnet (basnetbabin@gmail.com)

Tirtha Shrestha (Tirtha_48@yahoo.com.au)

Ghanshyam Kharel (Gskharel@gmail.com)

Rajeev Ojha (rajeевnet@hotmail.com)

Version: 1 Date: 24 Jun 2019

Author’s response to reviews:

Thank you for the valuable comments. We addressed all the technical and ethical comments raised by editor.

Similarly, language and some technical changes (in table 3 and p values in text) have been done as suggested by the reviewer.

However, in table 2, the contents of the table is such that it would not be appropriate to keep OR and CI in the same column. I hope this will not be an issue.

Regards,

Authors